Author's response to reviews

Title: A comparison of self reported air pollution problems and GIS-modeled levels of air pollution in people with and without chronic diseases

Authors:

Fredrik Niclas Piro (fnp@econ.no)
Christian Madsen (christian.madsen@fhi.no)
Oyvind Naess (oyvind.nass@medisin.uio.no)
Per Nafstad (per.nafstad@medisin.uio.no)
Bjorgulf Claussen (bjorgulf.claussen@medisin.uio.no)

Version: 5 Date: 23 February 2008

Author's response to reviews:

Thank you for the pre-acceptance of our manuscript. We have made some extremely minor changes in the manuscript, after carefully checked for typographical errors. In table 1, on pages 24-25, we have removed a comma in the top column (N=14,294). We have also in the column to the left (Variables) removed all commas, and replaced them with periods. This was in-consistent in the last version of the manuscript. We have not made any other changes and do not feel you have to check our manuscript again.

Kind regards, Fredrik Niclas Piro